Bariatric surgery, lifestyle interventions and orlistat for severe obesity: the REBALANCE mixed-methods systematic review and economic evaluation

Alison Avenell, 1* Clare Robertson, 1 Zoë Skea, 1 Elisabet Jacobsen, 2 Dwayne Boyers, 2 David Cooper, 1 Magaly Aceves-Martins, 1 Lise Retat, 3 Cynthia Fraser, 1 Paul Aveyard, 4 Fiona Stewart, 1 Graeme MacLennan, 1 Laura Webber, 3 Emily Corbould, 3 Benshuai Xu, 3 Abbygail Jaccard, 3 Bonnie Boyle, 1 Eilidh Duncan, 1 Michal Shimonovich 1 and Marijn de Bruin 5

Declared competing interests of authors: Alison Avenell, Clare Robertson and Graeme MacLennan acknowledge funding from the National Institute for Health Research Health Technology Assessment programme for projects outside this work.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published November 2018

DOI: 10.3310/hta22680

Plain English summary

Surgery, lifestyle interventions, orlistat for severe obesity

Health Technology Assessment 2018; Vol. 22: No. 68

DOI: 10.3310/hta22680

NIHR Journals Library www.journalslibrary.nihr.ac.uk

¹Health Services Research Unit, University of Aberdeen, Aberdeen, UK

²Health Economics Research Unit, University of Aberdeen, Aberdeen, UK

³UK Health Forum, London, UK

⁴Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

⁵Health Psychology, University of Aberdeen, Aberdeen, UK

^{*}Corresponding author a.avenell@abdn.ac.uk

Plain English summary

Severe obesity (body mass index of \geq 35 kg/m²) increases the risk of many illnesses. We reviewed the evidence for the acceptability, effectiveness and value for money of weight-management programmes (WMPs) for adults with severe obesity. We looked at surgery for obesity, diet and exercise WMPs and the drug orlistat.

We examined 131 trials, 26 UK studies, 33 studies of people's views and 46 studies of the value for money of WMPs. We undertook new research on the value for money of different weight-management approaches for the NHS.

Reviewed studies were often selective in who they included and did not follow people up for long enough.

Surgery for obesity had the best weight-loss results and could be a good use of NHS resources, compared with no surgery or WMPs. Of non-surgical approaches, very low-calorie diets (VLCDs) produced the best weight-loss result at 12 months, but it was unclear if this was sustained for longer. Adding a VLCD to an existing WMP was not shown to be a good use of NHS resources. However, most WMPs, including those with VLCDs, appeared to be a good use of NHS resources compared with no WMP.

The best results for long-term non-surgical weight loss over nearly 10 years came from a WMP with a low-fat reducing diet, a calorie goal of 1200–1800 kcal/day, initial meal replacements or meal plans, a tailored exercise programme, cognitive—behavioural therapy, intensive group and individual support, and follow-up by telephone or e-mail. This would be more costly for the NHS than simpler WMPs.

Low-carbohydrate Atkins-type diets, higher protein intakes or the use of meal replacements had small effects on improving weight loss at 12 months but had no longer-term effect. Increasing physical activity helped to prevent long-term weight regain, as did receiving longer-term help with diet or using orlistat.

Adding telephone or internet support, and group support, also helped to keep weight off. Participants in WMPs valued novelty in WMPs, WMPs endorsed by health-care providers and belonging to a group with people who shared similar issues.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.513

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 15/09/04. The contractual start date was in July 2016. The draft report began editorial review in July 2018 and was accepted for publication in October 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2018. This work was produced by Avenell et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk